Hospital at Home admission avoidance with comprehensive geriatric assessment to maintain living at home for people aged 65 years and over: a RCT

Sasha Shepperd,^{1*} Andrea Cradduck-Bamford,¹ Christopher Butler,² Graham Ellis,³ Mary Godfrey,^{4†} Alastair Gray,¹ Anthony Hemsley,⁵ Pradeep Khanna,⁶ Peter Langhorne,⁷ Petra Mäkelä,⁸ Sam Mort,² Scott Ramsay,⁹ Rebekah Schiff,¹⁰ Surya Singh,¹ Susan Smith,² David J Stott,⁷ Apostolos Tsiachristas,¹ Angela Wilkinson,¹¹ Ly-Mee Yu² and John Young¹²

Declared competing interests of authors: Sasha Shepperd reports membership of the National Institute for Health Research (NIHR) Health and Social Care Delivery Research programme Commissioned Prioritisation Committee (2015–18) during the conduct of this randomised trial and grants from the NIHR and the Medical Research Council. Chris Butler reports receiving grant funding from the NIHR as a NIHR Senior Investigator and grants from the NIHR Health Protection Research Unit on Health Care Associated Infections and Antimicrobial Resistance and the NIHR in his role as Director of the NIHR MedTech and In Vitro Diagnostics Cooperative for innovative diagnostic and monitoring technology to enhance Community Healthcare. He also held grants from the NIHR Health Technology

¹Nuffield Department of Population Health, University of Oxford, Oxford, UK

²Nuffield Department of Primary Care Health Sciences, University of Oxford, Oxford, UK

³School of Health and Life Sciences, Glasgow Caledonian University, Glasgow, UK

⁴Leeds Institute of Health Sciences, Academic Unit of Ageing and Stroke Research, Leeds, UK

⁵Royal Devon and Exeter NHS Foundation Trust, Exeter, UK

⁶Aneurin Bevan University Health Board, Newport, UK

⁷Institute of Cardiovascular and Medical Sciences, University of Glasgow, Glasgow, UK

⁸Department of Health Services Research and Policy, London School of Hygiene and Tropical Medicine, London, UK

⁹Department of Medicine for the Elderly, St John's Hospital, Livingston, UK

¹⁰Guy's and St Thomas' NHS Foundation Trust, London, UK

¹¹Medicine for the Elderly Department, Victoria Hospital, Kirkcaldy, UK

¹²Academic Unit of Elderly Care and Rehabilitation, University of Leeds, Leeds, UK

^{*}Corresponding author sasha.shepperd@ndph.ox.ac.uk †In memoriam

Assessment programme (reference 12/33/12) as well as other publicly funded grant giving bodies. He was also a member of the Efficacy and Mechanisms Evaluation Board from 2011 to 2016. He also reports personal fees from Roche Molecular Systems, grants and personal fees from Roche Molecular Diagnostics (Pleasanton, CA, USA), personal fees from Pfizer Inc. (Pfizer Inc., New York, NY, USA), and grants and personal fees from Janssen Pharmaceutical (Beerse, Belgium) outside the submitted work. Graham Ellis reports grants from NIHR during the conduct of the study. In addition, he is employed part time as a clinical advisor on ageing and health to NHS Scotland's (Edinburgh, UK) Chief Medical Officer and has advised the Scottish Cabinet Secretary for Health and Sport regarding implementation of Hospital at Home in Scotland. Alastair Gray reports grants from NIHR during the conduct of the study. Peter Langhorne reports that he was on the Health Technology Assessment CET Committee and Health Technology Assessment End of Life Care Add on Studies Committee until 2019. David J Stott reports grants from NIHR during the conduct of the study. Ly-Mee Yu was a panel member of the NIHR Health Technology Assessment Efficient Study Designs Committee and NIHR Research for Patient Benefit (2013–18), during the conduct of this research.

Published January 2022 DOI: 10.3310/HTAF1569

Plain English summary

Admission avoidance hospital at home with CGA Health and Social Care Delivery Research 2022; Vol. 10: No. 2 DOI: 10.3310/HTAF1569

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

What is the aim of this study?

Admission avoidance hospital at home is a service that is provided in some parts of the UK, and elsewhere, as an alternative to admission to hospital, but we do not know how this service compares with being admitted to hospital. The hospital-at-home services we evaluated were led by a geriatrician, provided multidisciplinary health care in a patient's home and gave access to hospital-based services and usual primary care services. The aim of this randomised trial was to find out if providing health care in hospital at home (sometimes called health care in the home), instead of in the hospital, helped older people to recover from a decline in their health. We also aimed to find out how much this type of health care costs, compared with hospital care, and how people experienced health care in the home and in hospital. Older people from nine locations across the UK who had experienced a change in their health and for whom health care in hospital had been considered, took part in this research. We collected data on various aspects of their health, including possible confusion, their ability to move around and to look after themselves and where they were living 6 and 12 months after the study had started.

Key messages

We found that hospital at home is another way to deliver health care to some older people who experience a decline in their health and are referred to hospital at home or hospital for a range of health problems. There was no difference in the numbers living at home at 6-month follow-up. We also found that hospital at home is less costly than health care in hospital for a certain group of older people, although it is not suitable for all older people who are unwell. Older people, whether they receive hospital at home or hospital-based health care, work hard to recover from being unwell. Their families and caregivers might provide substantial support to help them maintain their health, and this could include monitoring the patient while they are receiving health care in their home and helping to plan care following the patient's discharge from hospital at home or hospital.

Health and Social Care Delivery Research

ISSN 2755-0060 (Print)

ISSN 2755-0079 (Online)

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

This journal was previously published as *Health Services and Delivery Research* (Volumes 1–9); ISSN 2050-4349 (print), ISSN 2050-4357 (online)

The full HSDR archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hsdr. Print-on-demand copies can be purchased from the report pages of the NIHR Journals Library website: www.journalslibrary.nihr.ac.uk

Criteria for inclusion in the Health and Social Care Delivery Research journal

Reports are published in *Health and Social Care Delivery Research* (HSDR) if (1) they have resulted from work for the HSDR programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

HSDR programme

The HSDR programme funds research to produce evidence to impact on the quality, accessibility and organisation of health and social care services. This includes evaluations of how the NHS and social care might improve delivery of services.

For more information about the HSDR programme please visit the website at https://www.nihr.ac.uk/explore-nihr/funding-programmes/health-services-and-delivery-research.htm

This report

The research reported in this issue of the journal was funded by the HSDR programme or one of its preceding programmes as project number 12/209/66. The contractual start date was in July 2014. The final report began editorial review in February 2020 and was accepted for publication in October 2020. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HSDR editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, NETSCC, the HSDR programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, NETSCC, the HSDR programme or the Department of Health and Social Care.

Copyright © 2022 Shepperd *et al.* This work was produced by Shepperd *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Chair of HTA and EME Editorial Board and Editor-in-Chief of HTA and EME journals. Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Consultant Advisor, Wessex Institute, University of Southampton, UK

Ms Tara Lamont Senior Adviser, Wessex Institute, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, Wessex Institute, Faculty of Medicine, University of Southampton, UK

Dr Rob Riemsma Reviews Manager, Kleijnen Systematic Reviews Ltd, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk